Belief identification by proxy^{*}

Elias $Tsakas^{\dagger}$

Maastricht University

February, 2025

Abstract

It is well known that individual beliefs cannot be identified using traditional choice data, unless we exogenously assume state-independent utilities. In this paper, I propose a novel methodology that solves this long-standing identification problem in a simple way. This method relies on extending the state space by introducing a proxy, for which the agent has no stakes conditional on the original state space. The latter allows us to identify the agent's conditional beliefs about the proxy given each state realization, which in turn suffices for indirectly identifying her beliefs about the original state space. This approach is analogous to the one of instrumental variables in econometrics. Similarly to instrumental variables, the appeal of this method comes from the flexibility in selecting a proxy.

KEYWORDS: Belief identification, state-dependent utility, proxy, actual belief, axiomatic foundation.

JEL CODES: C81, C90, D80, D81, D82, D83.

1. Introduction

Identifying subjective beliefs is a central problem in economics that dates back to the seminal contributions of Ramsey (1931), De Finetti (1937) and Savage (1954). While this literature originally focused on providing foundations for subjective probability, more recently economists have also recognized the practical importance of the question (Manski, 2004). This renewed interest comes from the fact that beliefs are nowadays used for a broad range of purposes, e.g., to make out-of-context predictions; to obtain, compare, and aggregate forecasts; to study irregularities in information processing,

[†]Department of Microeconomics and Public Economics, Maastricht University, P.O. Box 616, 6200 MD, Maastricht, The Netherlands; Homepage: www.elias-tsakas.com; E-mail: e.tsakas@maastrichtuniversity.nl

^{*}I am grateful to editor Andrea Galeotti and three anonymous referees for their excellent comments which led to significant improvement of the paper. I am also greatly indebted to Edi Karni, Jay Lu, Andy Mackenzie and Peter Wakker for their instrumental input at various stages of this project. Also many thanks to Mohammed Abdellaoui, Larbi Alaoui, Jean Baccelli, Aurelien Baillon, Georgios Gerasimou, Itzhak Gilboa, Duarte Goncalves, Pierfrancesco Guarino, Jeanne Hagenbach, Peter Hammond, Andreas Kleiner, Hannes Leitgeb, Lasse Mononen, Antonio Penta, Ludvig Sinander, Lorenzo Maria Stanca, Lise Vesterlund, Sasha Vostroknutov, Roberto Weber, Lars Wittrock, Horst Zank and the audiences at FUR (Ghent), RUD (Northwestern), LOFT (Groningen), CRETA Conference (Warwick), Workshop on Preferences, bounded rationality, and strategic interactions (Barcelona), Workshop on strategic communication, bounded rationality and complexity (Cergy), Workshop on strategic communication (Graz), Conference on Rethinking Economic Theory (Athens), Epicenter Workshop (Maastricht), RCEA (online), GAEL Grenoble and internal seminars in Maastricht University for very useful comments.

etc. Therefore, accurately measuring actual beliefs is of utmost importance for applied research and policy.¹

The traditional methodology for belief identification relies on observing betting behavior: an agent's choices among acts are supposed to reveal her beliefs over the state space (Savage, 1954; Anscombe and Aumann, 1963; Wakker, 1989). However, there is a foundational caveat, known as the *identification problem* (e.g., Drèze, 1961, 1987; Fishburn, 1973; Karni et al., 1983):

The agent's beliefs can be identified through traditional choices over acts, only if we exogenously assume her utility function to be state-independent.

Let me illustrate the problem by means of an example.² Suppose that a man suffers from Guillain-Barré syndrom, a serious neurological condition that has left him completely paralyzed, and it is unclear whether he will recover from the disease within the next year. Suppose that we want to identify the subjective probability that his wife attaches to him recovering. Since she loves her husband very much, she values extra money more in a world where he has recovered, compared to a world where he is paralyzed. Hence, we cannot assume her utility over monetary payoffs to be state-independent. As a result, the relationship between her beliefs and her willingness to accept a bet is confounded by her relative utility for money between the two states. And since the latter is no longer normalized to 1 —as it would have been under the state-independence assumption— we cannot identify beliefs from her betting behavior.

This poses a serious problem, as in many economic applications utility functions are statedependent, either due to intrinsic preferences over states or due to unobservable state-dependent side payoffs. For example, state-dependent preferences commonly arise in insurance problems (Arrow, 1974; Cook and Graham, 1977; Drèze and Rustichini, 2004; Karni, 2008b), legal judgments (Andreoni, 1991; Feddersen and Pesendorfer, 1998; Tsakas, 2017), criminal behavior (Lochner, 2007), real estate decisions (Case et al., 2012), medical decisions (Pauker and Kassirer, 1975, 1980; Lu, 2019), health policy (Delavande, 2008), and in psychology when studying motivated beliefs (Kunda, 1990; Benabou, 2015) or ingroup favoritism (Everett et al., 2015). In all these cases, state-dependent preferences will likely induce errors in belief measurement. Such errors are traditionally overlooked, not because they are not necessarily significant or important, but rather because they cannot be separated from systematic components of the agent's beliefs with the traditional elicitation tools.

Take for instance the extensive literature on motivated beliefs, where the main hypothesis is that intrinsic preferences over the state space lead to systematically biased beliefs in the direction of the preferred state (Kunda, 1990; Benabou, 2015). For instance, people reportedly overestimate their own skill or performance (Svenson, 1981; Eil and Rao, 2011; Zimmermann, 2020; Möbius et al., 2022) or the chances of political events that they like (Bullock et al., 2015; Thaler, 2024). Such overconfidence has important downstream consequences in behavior, e.g., in a political setting, it affects voter turnout and political polarization (Ortoleva and Snowberg, 2015). In order to unbiasedly estimate the effect of overconfidence on behavior, it is important to separate belief distortions due to motivation from possible measurement errors due to falsely assuming state-independent utilities.

Another instance of measurement errors due to state-dependent preferences arises in the surging literature that uses reported beliefs in order to predict behavior in different settings, e.g., in finance (Gennaioli et al., 2015), health (Delavande, 2008) and crime (Lochner, 2007), just to mention a few.

¹Throughout the paper I assume that an actual belief exists, and it is therefore interpreted as an unobservable primitive. Still, my entire analysis can be easily adapted to classical settings where the belief is interpreted as a parameter that only acquires meaning within a SEU model (Section 6).

²This is a modification of the standard example, which was originally used in a letter correspondence between Bob Aumann and Leonard Savage (Drèze, 1987), and has been used widely in this literature since then. We will use this story as our running example throughout the paper. For a formal presentation, see Example 1.

For instance, the subjective probability that women attach to getting pregnant, combined with their choices among birth control methods, allow us to identify their (actual) utilities over pregnancy outcomes (Delavande, 2008), and subsequently to predict women's response to different policy interventions. However, all this presupposes that we elicit women's beliefs without measurement errors, which is rather difficult to do with the usual methods due to the fact that women have obvious stakes in whether they get pregnant or not.

As a final instance of measurement errors because of state-dependent preferences, consider cases where decision makers use experts' expectations as input for their own choices. For instance, small investors will often adopt the forecasts of institutional investors before making their own portfolio allocation choices; occasional gamblers will adopt the forecasts of professional gamblers before deciding where they will place their money; prospective home owner will adopt the forecasts of real estate investors before bidding for a house. However, in practice, all these decision makers rarely take into account that there are usually unobservable state-dependent side payments for these experts, e.g., in the form of an unobservable existing portfolios for the institutional investors or the real estate investors, and similarly in the form of unobservable arbitrage opportunities for the professional gambler (Vlastakis et al., 2009). In the presence of such side payments, the measurement error will likely arise as a consequence of hidden hedging opportunities (Blanco et al., 2010).

Historically, the identification problem was already noticed during the very early days of decision theory, but was consciously put aside: as Leonard Savage admits in his well-known letter correspondence with Bob Aumann, "the problem is serious, but I am willing to live with it until something better comes along" (Drèze, 1987). Most of the proposed solutions rely on non-traditional choice data (see literature overview below). However, none of them is unanimously accepted as the standard one, largely because they require complex and cumbersome data. This suggests that the identification problem is both hard-to-solve and still open.

In this paper, I propose a novel solution, which is both theoretically sound and tractable. The key idea is to stick to the standard methodology of using traditional choice data, albeit over an extended state space. I do this by taking a product space, where one dimension is the original state space and the second dimension is what I call a *proxy*. The crucial feature is that the agent does not have any stakes in the realization of the proxy conditional on our original state space. This allows us to identify her conditional beliefs about the proxy using standard elicitation mechanisms. The main result leverages these conditional beliefs about the proxy to uniquely identify the actual belief about the original state space (Theorem 1).

Let me illustrate the identification result in the context of the running example. Suppose that there is a promising new experimental drug in the market which is believed to expedite the recovery time from Guillain-Barré. The husband is eligible to participate in the last phase of the clinical trial. A proxy describes the two possible contingencies regarding his participation, i.e., he will receive either the drug (t_1) or a placebo (t_2) . The wife is informed that his chances of being placed in the treatment group are 50%, but she is not told in which group he is actually placed in the end. Of course, if she were to learn that he received the drug, her belief would change to some $\nu := \pi_S(\cdot|t_1)$; on the other hand, if she learned that he received the placebo, her belief would remain unchanged at $\mu := \pi_S(\cdot|t_2)$.³ For a graphical illustration, see Figure 1 below.

The crucial feature is that the wife *does not have any stakes in the realization of the proxy conditional on her husband's health status.* For instance, conditional on him recovering (resp., remaining paralyzed), she will not care about whether he has taken the drug or the placebo, and therefore her

³Here, I implicitly assume that the husband does not even know that he is participating in the clinical trial, and therefore his health cannot be affected by psychological factors, such as the well-known placebo effect.



Figure 1: The two contingencies that the proxy describes are "the husband takes the experimental drug (t_1) " and "the husband takes the placebo (t_2) ". The prior probability of the husband (randomly) being placed in the control group is 50%. Conditional on the placebo being taken, the wife's belief remains at μ . Conditional on the drug being taken, her belief changes to ν .

utility from money will not depend on the realization of the proxy.⁴ Hence, we can identify her conditional beliefs over T given that her husband has recovered (viz., $\pi_T(\cdot|s_1)$), as well her conditional beliefs over T given that he has remained paralyzed (viz., $\pi_T(\cdot|s_2)$). In both cases, we can do this, using standard elicitation tasks, without worrying about the identification problem (Section 4).

Then, as long as S and T are not independent, we can uniquely identify her conditional belief $\pi_S(\cdot|t_2)$ given that he has taken the placebo, which is the belief μ she would have held if we had not introduced the proxy. A generalization of this simple example to any finite state space and any suitable proxy constitutes my main identification result (Theorem 1).

Notably, my identification approach bears a striking conceptual similarity with the use of instrumental variables in econometrics. Recall that in IV regression we replace one awkward exogenous assumption (viz., orthogonality) with another exogenous assumption which is easier to justify (viz., the exclusion restriction). In the same way, instead of exogenously assuming that the wife does not care about her husband's health status, we assume that she does not care about his treatment conditional on his eventual health status, with the latter being much easier to defend than the former. Thus, in both settings, it is not that we stop imposing exogenous assumptions all together, but rather that we strategically select the domain (viz., the instrumental variable and the proxy, respectively) so that the exogenous assumptions can be justified on the basis of common sense and/or existing literature.

The flexibility to choose the instrumental variable is what has made IV regression so popular. Likewise, the flexibility in choosing a proxy from a large pool of potential candidates, is what makes belief identification by proxy an appealing method. In Section 7, I show that it is indeed essentially always possible to construct a suitable proxy, even in cases where we cannot easily pick one off the shelf. Then, I go on to revisit some applications where identification by proxy corrects the belief measurement error which is likely to arise due to falsely assuming state-independent utilities.

The existing literature is roughly split in two streams, one that aims at providing tools for belief elicitation purposes, and an axiomatic one that focuses on achieving identification theoretically. A major conceptual difference between these two streams is that the former explicitly assumes that beliefs are an actual primitive (similarly to this paper), whereas the latter treats beliefs as a parameter within a model without much concern on whether these are actual beliefs or not. For an overview, see Drèze and Rustichini (2004), Grant and van Zandt (2008), Karni (2008a, 2014) and Baccelli (2017).

Starting with the first stream, the only papers that introduce mechanisms for eliciting beliefs under state-dependent preferences are Karni (1999) and Jaffray and Karni (1999), with the latter

⁴The implicit assumption that I make for the sake of this simple example is that the drug is known not have any side-effects that would make the wife care about whether her husband has taken it or not conditional on eventual health condition.

proposing two different mechanisms. In particular, Karni (1999) and the first mechanism of Jaffray and Karni (1999) rely on assuming bounded state utilities, and they approximate the actual beliefs as monetary incentives grow arbitrarily large. The problem is that this approach requires either extreme experimental costs, or hypothetical data. These pitfalls are recognized by the authors, who point out that in those early days there was no other option (e.g., Karni, 1999, p.485). The second mechanism in Jaffray and Karni (1999) assumes state-dependence in the form of unobserved state-dependent side payments: first it proceeds to elicit these payments, and subsequently to elicit beliefs using standard techniques. Unfortunately, this is a rather restrictive setting: in many applications, preferences over states are intrinsic. Moreover, eliciting the state-dependent payments is data-demanding.

Turning to the second stream, the various attempts within axiomatic decision theory differ in terms of the non-traditional choice domain they consider. There are two main methodological approaches, both relying on the agent's beliefs being somehow revised at some instance.

The first such method, which was mainly followed in the early days, is based on exogenously manipulating the context and a fortiori the agent's beliefs. For instance, Fishburn (1973) allows for comparison between acts conditional on different events. Karni et al. (1983), Hammond (1999) and Karni and Schmeidler (2016) introduce hypothetical preferences over acts conditional on exogenously given probabilities over the states. Schervish et al. (1990) allow the agent to compare lotteries at different states. Karni (1992, 1993a) allows the analyst to observe preferences conditional on different events. In some way, all of these papers assume the analyst to observe the agent's hypothetical choices. This feature makes the implementation of these methods cumbersome.

The second method relies on the idea that the agent herself can affect the state realization, and is called the *moral hazard approach*. It originally appeared in the early sixties (Drèze, 1961), before resurfacing (Drèze, 1987; Drèze and Rustichini, 1999) and recently receiving attention again (Baccelli, 2021). This literature relates to my work on a high level. The difference is that in moral hazard the analyst relies on the agent influencing the state realization, whereas in my case it is sometimes the analyst who may choose to influence the state realization (Example 4). The second major difference is that the moral hazard methodology applies only to settings where the state has not been realized yet, while my method also applies to factual beliefs.

Closer to my paper is the recent work of Lu (2019), which leverages stochastic choice data in order to identify the agent's beliefs. This approach relies on observing choices conditional on (many) independent realizations of a given signal. It is similar in spirit to mine, in that it also extends the state space by considering a second dimension (i.e., the set of signal realizations). Its main drawback is that it requires a large dataset of independent observations, which is not always feasible, e.g., stochastic choice data cannot be collected when uncertainty is about a one-shot event, like for instance the husband's health in our running example, or the outcome of the upcoming presidential elections. I further elaborate on the relationship between my work and the one of Lu (2019) in Section 3.

Between the moral hazard approach and the aforementioned stochastic choice approach, one can place a sequence of papers that rely on the agent influencing the state realization and choosing an act, conditional on different signals (Karni, 2011a,b, 2014). The common element across these papers of Karni, the paper of Lu, and my work is that all can be thought to rely on traditional choice data over an extended state space. Nevertheless, the way this general idea is implemented is very different.

There is also related work on the identification problem within non-expected utility models: While the problem is resolved in some cases outside SEU (Chew and Wang, 2020; Mononen, 2023), in other cases it persists (Karni, 2020).

Finally, my work is methodologically related to a recent paper by Alaoui and Penta (2024), whose aim is to separately identify risk preferences from utility from wealth. Their identification strategy relies on introducing an additional dimension in the form of a yardstick, which plays a similar role to the one that the proxy plays in my paper.

Section 2 presents the background and formally introduces the identification problem. Section 3 presents my method and formally states my main identification result. Section 4 relates my approach with the standard belief elicitation mechanisms. Section 5 leverages my identification result to provide a well-founded definition of actual utility function. Section 6 provides decision-theoretic foundations for a proxy, thus formally distinguishing between exogenous and testable assumptions. Section 7 elaborates into practical aspects of my approach. Section 8 concludes. All proofs are relegated to the Appendix.

2. The identification problem formalized

Consider a finite state space $S = \{s_1, \ldots, s_K\}$. A (female) agent has a full-support belief $\mu \in \Delta(S)$, called the *actual belief*, which we want to identify.

An act is a function $f : S \to Q$ that maps each state $s \in S$ to a consequence $f_s := f(s)$ in a convex subset Q of a finitely dimensional Euclidean space. The set of all acts is denoted by \mathcal{F}_S . Observed choices among acts are henceforth called traditional choice data.⁵ Suppose that the agent's choices are consistent with a preference relation \succeq over \mathcal{F}_S that admits a *State-Dependent Subjective Expected Utility (abbrev., SEU)* representation. That is, there exists some state-dependent utility function $u : S \times Q \to \mathbb{R}$ such that, the preference relation \succeq is represented by the SEU function

$$\mathbb{E}_{\mu}(u(f)) := \sum_{s \in S} \mu(s) u_s(f_s), \tag{1}$$

where $u_s(q) := u(s,q)$ is the state-utility from outcome q at state s. Formally, this means that for any two acts $f, g \in \mathcal{F}_S$, the following equivalence holds:

$$f \ge g \iff \mathbb{E}_{\mu}(u(f)) \ge \mathbb{E}_{\mu}(u(g)).$$
 (2)

Whenever this is the case, we will say that the pair (u, μ) represents \geq .

The appeal of SEU is that it allows us to separate beliefs from utilities. Is this enough for identifying the agent's actual beliefs? Unfortunately, not! The reason is that the pair (u, μ) is not the only SEU representation. Namely, take any other full-support belief $\tilde{\mu} \in \Delta(S)$, and define the rescaled utility function $\tilde{u}: S \times Q \to \mathbb{R}$ so that, for every $s \in S$, it is the case that

$$\tilde{u}_s = \alpha_s + \beta \frac{\mu(s)}{\tilde{\mu}(s)} u_s,\tag{3}$$

where $a_s \in \mathbb{R}$ and $\beta > 0$. Then, for every $f \in \mathcal{F}_S$, we obtain

$$\mathbb{E}_{\tilde{\mu}}(\tilde{u}(f)) = \alpha + \beta \mathbb{E}_{\mu}(u(f)), \tag{4}$$

where $\alpha := \sum_{s \in S} \alpha_s \tilde{\mu}(s)$. Hence, the pair $(\tilde{u}, \tilde{\mu})$ constitutes an alternative SEU representation of the same preferences. So, even if we had access to the complete preference relation (e.g., by observing data from a complete experiment), we would not be able to tell if the actual belief is μ or $\tilde{\mu}$, i.e., beliefs cannot be identified from choices over \mathcal{F}_S . This is known as the *identification problem* of SEU.

⁵Formally, traditional choice data take the form of a collection of menus $F \subseteq \mathcal{F}_S$ (i.e., the experiment), together with a choice correspondence that associates each menu F in this experiment with a non-empty set of observed choices $C(F) \subseteq F$ (i.e., the data). The experiment is called complete whenever it contains every compact subset of \mathcal{F}_S .

The early solution to the identification problem was to essentially assume it away. This approach boils down to selecting a *State-Independent Subjective Expected Utility (abbrev., SI-SEU)* representation $(\bar{u}, \bar{\mu})$ among the infinitely many SEU representations, i.e., one such that $\bar{u}_s(q) = \bar{u}(q)$ for all $s \in S$ and all $q \in Q$. It is well-known that all SI-SEU representations involve the same belief $\bar{\mu}$ (Savage, 1954; Anscombe and Aumann, 1963; Wakker, 1989). Then, according to this early solution, this belief $\bar{\mu}$ is labelled as the actual belief.⁶ This is known as the assumption of state-independent utilities. Unfortunately, as illustrated in the following example, this assumption corresponds to an arbitrary model selection.

Example 1. (THE WIFE'S PROBLEM). Recall the introductory example, in which a man suffers from Guillain-Barré syndrom, and it is unclear whether he will recover (state s_1) or remain paralyzed (state s_2). Suppose that his wife is a risk-neutral SEU maximizer, i.e., she evaluates monetary payoffs, at states s_1 and s_2 respectively, using the state-utility functions

$$u_1(q) = \gamma_1 q$$
 and $u_2(q) = \gamma_2 q$,

with $\gamma_1, \gamma_2 > 0$. So, when we observe her being indifferent between buying and not buying an insurance with payout of \$100k and premium of \$10k, we conclude that her beliefs satisfy

$$\frac{\mu(s_2)}{\mu(s_1)} = \frac{\gamma_1}{9\gamma_2}.\tag{5}$$

Obviously, in order to identify her actual belief, we must specify the parameters γ_1 and γ_2 . Stateindependent utilities correspond to one such specification, which sets $\gamma_1 = \gamma_2$, and subsequently yields $\bar{\mu}(s_1) = 0.90$. However, this is just one of the infinitely many parameter specifications that satisfy Equation (5), and in this sense it is an arbitrary choice.

To complicate things even further, oftentimes a SI-SEU representation does not even exist in the first place. This is for instance the case when state-preferences over consequences differ across states, either in some inherent way (Example 2), or because of unobservable state-contingent side payments (Example 3).

Example 2. Continuing with the same example, suppose that the wife's preferences were instead represented by the SEU representation (u, μ) , where

$$u_1(q) = q^2$$
 and $u_2(q) = q$.

The idea is that the wife is risk-seeking whenever her husband is healthy. In order for a SI-SEU representation to exist, it must necessarily be the case that $u_1 = \alpha + \beta u_2$ for some $\alpha \in \mathbb{R}$ and $\beta > 0$. It is not difficult to verify that such parameters do not exist.

Example 3. In the running example, suppose that the wife's preferences were instead represented by the SEU representation (u, μ) , where

$$u_1(q) = q^2$$
 and $u_2(q) = (c+q)^2$.

The interpretation is that the wife is risk-seeking with quadratic state-utility function at both states, and has an unobservable side payoff at s_2 , e.g., the wife has already bought another insurance that we are not aware of. Once again, there are no parameters $\alpha \in \mathbb{R}$ and $\beta > 0$ such that $u_1 = \alpha + \beta u_2$. So again, there is no SI-SEU representation.

⁶Strictly speaking, the same belief $\bar{\mu}$ would have been uniquely identified even if we had started with a quasi state-independent utility function, $\bar{u}_s(q) = \alpha_s + \bar{u}(q)$, instead. The crucial property of such a representation is that the state-independent and the state-dependent part of the utility function are linearly separable. This is because, by (3) and (4), adding a state-dependent constant does not affect the belief in the SEU representation. I thank an anonymous referee for pointing out this subtle issue. I further elaborate in Section 4.

Whenever there is no SI-SEU representation, the identification problem becomes immediately visible, as there is no obvious candidate among the infinitely many SEU representations to exogenously assume. However, this does not mean that in the converse scenario where a SI-SEU representation exists, the identification problem is resolved. In fact, the existence of a SI-SEU representation and the identification problem are conceptually orthogonal: the former is about axioms of choice being satisfied, while the latter is about model specification, conditional on these axioms being satisfied. So, why is there a widespread misconception that they are linked? The reason is twofold and can be traced back into the history of SEU.

First of all, in the early days of decision theory, the literature was not interested in identifying actual beliefs. Instead, the focus was on predicting choices among acts in \mathcal{F}_S , and on establishing that there exists a well-founded definition of subjective probability. And since SI-SEU representations served both purposes, the theoretical literature started treating $(\bar{u}, \bar{\mu})$ as the default model, whenever it existed. Thus, when the applied literature eventually started being interested in identifying actual beliefs, it simply adopted $\bar{\mu}$ as 'the correct definition of actual belief'. This is highlighted in two facts: first, (almost) the entire belief elicitation literature consists of mechanisms that elicit $\bar{\mu}$ (see Section 4 for an overview of this literature); and second, the entire literature on State-Dependent SEU is motivated by the non-existence of SI-SEU representation (Drèze and Rustichini, 2004; Grant and van Zandt, 2008; Karni, 2008a, 2014; Baccelli, 2017, and references therein).

The second source of the misconception comes from the fact that state-independent utilities are often confused terminologically with state-independent preferences, with the latter being typically captured by axioms such as the state-monotonicity axiom in Anscombe and Aumann (1963), or P3-P4 in Savage (1954), or state-independent preference intensity in Wakker (1989). However, state-independent preferences are weaker than state-independent utilities, as the former simply guarantees the existence of a SI-SEU representation, whereas the latter selects a SI-SEU representation as the 'correct model' whenever one exists.

But how big of a problem is it to assume state-independent utilities (assuming of course that a SI-SEU representation exists)? The answer depends on the purpose of belief identification. If we simply want to predict the wife's insurance choices in Example 1, it is no big deal to assume state-independent utilities, as all SEU models (including SI-SEU) deliver the exact same predictions. If, on the other hand, we want to use the wife's beliefs for other purposes (e.g., as data in an experiment that studies motivated beliefs, or for predicting out-of-context if she will book a vacation for next summer), we need to select a SEU representation that involves her actual belief. In this last case, it is reasonable to assume state-independent utilities only if we are confident that the agent has no stakes over the state realization. We further elaborate on this point in Section 7.2, where we present several economic applications where the literature naively assumes state-independent utilities.

Remark 1. Regarding terminology, throughout the paper, "having no stakes in the state realization" is used synonymously to "the utility function being state-independent". As a side remark, Ramsey (1931) referred to this case as one where all relevant events in S are "ethically neutral".

3. Main identification result

In this paper, I propose a novel approach that solves the identification problem in a theoretically sound and practically tractable way. The idea is to continue using traditional choice data, albeit over an appropriately extended state space. Formally, I introduce another state space $T = \{t_1, \ldots, t_N\}$, and define the product space $S \times T$. The agent's actual belief over the extended state space is denoted by $\pi \in \Delta(S \times T)$. For any nonempty $A \subseteq S$ and $E \subseteq T$, we respectively denote the marginal conditional beliefs $\pi_T(E|A) := \pi(A \times E|A \times T)$ and $\pi_S(A|E) := \pi(A \times E|S \times E)$. Marginal (unconditional) beliefs are simply denoted by π_T and π_S respectively.

Definition 1. (PROXY). We say that T is a proxy for S whenever the following are satisfied:

 (P_0) CONDITIONAL NO STAKES: The agent has no stakes in T conditional on the realization of S.

(P₁) UNINFORMATIVE EVENT: There is some event $E \subseteq T$ such that $\pi_S(\cdot|E) = \mu$.

- (P₂) OBJECTIVE MARGINAL BELIEF: π_T agrees with an objective probability measure π_T^{obj} .
- (P₃) LINEAR INDEPENDENCE: $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$ are linearly independent in \mathbb{R}^T .

Condition (P_0) is the key feature of T. Formally, this is satisfied if and only if any incentivecompatible mechanism will elicit the agent's actual conditional beliefs $\pi_T(\cdot|s)$ for all $s \in S$ via the strategy method (see Section 4). Of course, ex ante the agent may still care about the realization of T, due to the correlation with S. But once uncertainty about S has been resolved, she has no longer any stakes in the realization of the proxy.

Condition (P_1) states that conditioning with respect to $E \subseteq T$ does not provide any information about S in comparison to the benchmark case where T is not introduced in the first place. Thus, E is called the uninformative event. Notice that the event E is not necessarily equal to T itself, meaning that the actual belief μ does not necessarily coincide with the unconditional marginal π_S (Examples 4 and 6 below).

Condition (P_2) consists of two parts. First, uncertainty about T is described by an objective marginal belief, π_T^{obj} , which is formed based on commonly known facts; second, it is assumed that the agent's actual marginal belief π_T agrees with the objective belief. This assumption often reflects the idea that π_T^{obj} has been publicly announced by the analyst (Examples 4 and 6 below), while in other cases it describes some demographic characteristic whose distribution is already commonly known to be π_T^{obj} (Example 5 below).

Condition (P_3) is perhaps the least-obvious of the four conditions. Loosely speaking, it guarantees that there is no redundant information within T. Whenever S is binary, this condition reduces to $\pi_T(\cdot|s_1) \neq \pi_T(\cdot|s_2)$, i.e. S and T are not independent.

Let me provide some examples for different types of proxies, while also illustrating the four conditions of Definition 1.

Example 4. (NEW DRUG). Recall the wife's problem, and similarly to the introduction suppose that there is a promising new experimental drug which supposedly helps recovery. The husband is eligible to participate in a clinical trial. The proxy describes the two possible groups in which the husband can be placed:

 $T = \{ \text{treatment group } (t_1), \text{ control group } (t_2) \}.$

Then, I argue that $(P_0) - (P_3)$ satisfy common sense:

- P_0 : The wife will not care if he has taken the drug or the placebo, once the husband's health condition is known.
- P_1 : The wife believes that in case he receives the placebo, his chance of recovery will remain unchanged, i.e., $\pi_S(\cdot|t_2) = \mu$. Hence, $\{t_2\}$ is the uninformative event.
- P_2 : The wife knows that the chance of him being included in the treatment group is 50%.
- P_3 : The wife believes that the drug will affect his recovery probability, i.e., $\pi_T(\cdot|s_1) \neq \pi_T(\cdot|s_2)$.

Such proxies are called *influential actions* (Tsakas, 2020), as the analyst (stochastically) influences the state realization via the proxy. \triangleleft

Example 5. (RELEVANT GENE). We consider the wife's problem once again. It is commonly known that recovery is correlated with the presence of a specific gene that 50% of all males carry. The gene has no other consequence. The wife does not know if her husband carries it or not. So, the proxy becomes:

 $T = \{ \text{gene } (t_1), \text{ no gene } (t_2) \}.$

Once again, I argue that $(P_0) - (P_3)$ satisfy common sense:

- P_0 : Conditional on the husband's health condition, the wife does not care if he has the gene.
- P_1 : Her actual belief is the one she holds before learning whether he has the gene or not, i.e., $\mu = \pi_T(t_1)\pi_S(\cdot|t_1) + \pi_T(t_2)\pi_S(\cdot|t_2)$. Hence, T is the uninformative event.
- P_2 : The wife knows that the probability of the husband having the gene is 50%.

 P_3 : The wife knows that recovery is correlated with the gene, i.e., $\pi_T(\cdot|s_1) \neq \pi_T(\cdot|s_2)$.

Such proxies are induced by a *demographic characteristic* whose distribution is commonly known. \triangleleft

Example 6. (EXPERT VERSUS CHARLATAN). We consider the wife's problem for a third time, supposing now that the wife is told that her husband's file was examined by some doctor who subsequently predicted that he will recover. The proxy describes the two possible expertise levels of the doctor:

$$T = \{ \text{expert } (t_1), \text{ charlatan } (t_2) \}.$$

Then again, I argue that $(P_0) - (P_3)$ are reasonable conditions:

- P_0 : Conditional on the husband's health condition, the wife will not care if the diagnosis has come from the expert or the charlatan.
- P_1 : Conditional on the diagnosis coming from a charlatan, the wife does not revise her beliefs, i.e., $\pi_S(\cdot|t_2) = \mu$. Hence, $\{t_2\}$ is the uninformative event.
- P_2 : The wife knows that the probability of the doctor being an expert is 50%.
- P_3 : The wife believes that the probability that the good news actually came from the expert is larger if the husband recovers than if he does not, i.e., $\pi_T(\cdot|s_1) \neq \pi_T(\cdot|s_2)$.

Such proxies are called *evidence with stochastic reliability*. A somewhat similar idea has been used in a different context by Thaler (2024).

Note that $(P_0) - (P_2)$ are all exogenous assumptions, in the sense that they cannot be corroborated with traditional choice data. In particular, (P_0) is exogenous for the same reason the assumption of state-independent utilities across S is also exogenous, i.e., we can only justify it on the basis of accepting that, upon fixing any $s \in S$, the agent has no intrinsic preference over realizations of T, nor does she have any unobservable side payments that depend on T. Condition (P_1) is also exogenous, as it makes direct reference to the actual belief μ , which is of course unidentifiable with traditional choice data. Finally, (P_2) is exogenous because it is an assumption regarding the unconditional belief about T, which is in turn a function of μ , as S and T are correlated. Such correlation also explains why we cannot directly elicit π_T , instead of exogenously assuming it to be equal to π_T^{obj} .

On the other hand, once we have assumed that (P_0) holds, (P_3) can be tested with traditional choice data. This is because under (P_0) , the conditional beliefs $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$ can be directly identified (Section 4), and once these are known it can be simply checked whether they are linearly independent. The formal distinction between exogenous and testable conditions is further discussed in Section 6, where I provide decision-theoretic foundations.

Then, my main result shows that we can leverage conditional beliefs about the proxy to identify the actual belief about S.

Theorem 1 (Main identification result). Assume that T satisfies $(P_0) - (P_2)$. Then, the actual belief $\mu \in \Delta(S)$ is uniquely identified if and only if (P_3) is satisfied.

The crucial step towards identification is to pin down the joint belief π . To do this, first, we identify the conditional beliefs $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$, leveraging the fact that the agent has no stakes in T conditional on S. Then, we mix these conditional beliefs with π_S , and we calibrate the marginal π_T against the objective marginal π_T^{obj} , similarly in spirit to the idea of Anscombe and Aumann (1963) who calibrate subjective probability distributions against objective ones.⁷ This calibration exercise allows us to identify π_S , and subsequently to obtain the joint belief π . Finally, by conditioning with respect to the uninformative event $E \subseteq T$ we obtain μ . Let me provide an illustration.

Example 4 (continued). Suppose that the wife reports probability $\pi_T(t_1|s_1) = 0.80$ of the husband receiving the drug conditional on recovering, and probability $\pi_T(t_1|s_2) = 0.40$ of the husband receiving the drug conditional on remaining paralyzed. There is a unique convex combination of the two aforementioned conditional probabilities inducing the objectively known $\pi_T(t_1) = 0.50$, i.e., we have $0.25\pi_T(\cdot|s_1) + 0.75\pi_T(\cdot|s_2) = \pi_T$. So, it is necessarily the case that $\pi_S(s_1) = 0.25$, meaning that the joint belief π is given by the following table:

| | recovers (s_1) | paralyzed (s_2) |
|-----------------|------------------|-------------------|
| drug (t_1) | 0.20 | 0.30 |
| placebo (t_2) | 0.05 | 0.45 |

Hence, her actual belief attaches probability $\mu(s_1) = \pi_S(s_1|t_2) = 0.10$ to her husband recovering. Note that the identification would not have been possible if $\pi_T(\cdot|s_1) = \pi_T(\cdot|s_2)$ (i.e., if S and T were independent), as in this case we would have obtained $\lambda \pi_T(\cdot|s_1) + (1 - \lambda)\pi_T(\cdot|s_2) = \pi_T$ for every $\lambda \in (0, 1)$.

Importantly, for my identification strategy to work, it must be the case that $|\operatorname{supp}(\pi_T)| \geq |S|$. Otherwise, (P_3) will be directly violated, since $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$ will not be linearly independent, and a fortiori we will not be able to identify the joint belief π . The good news is that the amount of data that we need is still quite small, e.g., whenever the state space is binary, we only need to elicit two probabilities.

Now that my identification method is clear, let me go back to the relationship with the one of Lu (2019). Technically, the two approaches are similar in that both introduce an additional dimension T, and subsequently take as input the conditional distributions $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$ and the marginal π_T , in order to obtain the joint distribution π . Finally both papers eventually condition with an uninformative event, which in Lu's case is always the entire set T. At the same time, the two approaches crucially differ in terms of where these inputs come from. In Lu (2019), the marginal distribution is approximated by the frequency of observed choices, while the conditional distributions are either exogenously given (when a single signal is used) or inferred from the data (when two ordered signals are used). On the other hand, in my paper things are upside down, i.e., the conditional distributions are directly elicited and the marginal is exogenously given. These differences are natural, given that Lu (2019) focuses on a specific set of applications, where a large number of choices is readily available, conditional on independent signal realizations from T. Such applications include hiring decisions, loan approvals, medical advice. However, this requirement rules out many cases where we want to elicit the agent's belief about a one-shot realization of S, e.g., if

⁷I would like to thank an anonymous referee for pointing out this connection.

we are interested in the wife's belief in our running example, or in some partian's belief about the winner of the upcoming presidential election.

On a high level, my approach bears a striking conceptual similarity to the use of instrumental variables in regression analysis. In particular, instead of automatically imposing an exogenous assumption (viz., orthogonality) on their variable of interest (viz., explanatory variable), econometricians assume it on another carefully-chosen variable (viz., instrumental variable). Similarly, instead of exogenously assuming without much thought that the agent has no stakes about the original state space (viz., S), I exogenously assume $(P_0) - (P_2)$ about a carefully-chosen proxy (viz., T). That is, in both cases, it is not that we stop imposing exogenous assumptions all together, but rather we only do so in specifically-chosen domains where the exogenous assumptions can be justified on the basis of common sense and/or well-established theoretical insights.

Such flexibility is what has made instrumental variables so successful. Similarly, I argue that my belief identification approach via proxies is appealing because there exists abundance of potentially suitable proxies, one of which will eventually satisfy common sense and/or will be consistent with well-established insights. That is, although $(P_0) - (P_2)$ might seem demanding ex ante, this is not really a problem, as we only need to find one proxy for which we will be comfortable with imposing these assumptions. For example, in case we find it difficult to justify (P_1) in Example 4 because of the placebo effect, we can switch to Example 5 where we are more comfortable with assuming that the wife does not have any information on the group of men that carries the gene. And at the end of the day, even if none of these readily available proxies satisfies us, we can always construct one specifically for the purpose of our identification exercise (Section 7).

4. Belief elicitation mechanisms

The common practice for identifying actual beliefs is by using belief elicitation mechanisms. These are direct mechanisms that incentivize the agent to reveal her actual belief (about the original state space S), by rewarding accurate reports and punishing inaccurate ones. A mechanism is formally written as a function

$$\sigma: \Delta(S) \to \mathcal{F}_S,$$

that takes as input the agent's reported belief $r \in \Delta(S)$, and returns as output a consequence $\sigma^r(s) \in Q$ upon state $s \in S$ being realized.

An elicitation mechanism is strictly incentive-compatible if truthful revelation of the actual belief is the agent's only optimal report, i.e., formally, for any actual belief $\mu \in \Delta(S)$, act $\sigma^{\mu} \in \mathcal{F}_S$ which is associated with reporting truthfully is strictly preferred to act $\sigma^r \in \mathcal{F}_S$ which is associated with any other report $r \neq \mu$. Existing mechanisms within the literature differ in how the incentives are framed.

The most popular belief elicitation mechanism is the quadratic scoring rule (De Finetti, 1937; Brier, 1950; Good, 1952). This is a direct mechanism that rewards the agent with a flat payment minus an inaccuracy penalty which is proportional to the squared (Euclidean) distance between the submitted report and the realized state. Formally, if the agent reports r and state s is realized, she will be paid the monetary amount

$$\sigma^{r}(s) = \alpha - \beta \|r - \mathbf{1}_{s}\|^{2}, \tag{6}$$

where $\beta > 0$ and $\mathbf{1}_s$ is the Dirac measure that puts probability 1 to state s.

The widely-accepted claim is that —whenever the agent is risk-neutral— quadratic scoring rules

are strictly incentive-compatible.⁸ Unfortunately, there is a subtle misconception which goes back to the belief identification problem, viz., without the assumption of state-independent utilities, the quadratic scoring rule is not incentive-compatible, even if the agent is risk-neutral.

Example 7. (QUADRATIC SCORING RULE IN THE WIFE'S PROBLEM). Recall the wife's problem from Example 1, and suppose that the wife is asked to report a probability $r(s_1) \in [0, 1]$ of her husband recovering. In return, a quadratic scoring rule will determine her payment according to formula (6). Now, suppose that the pair (u, μ) is a SEU representation which involves her actual belief μ . Then, the wife will submit a report that maximizes the following function:

$$\mathbb{E}_{\mu}(u(\sigma^{r})) = \mu(s_{1})\gamma_{1}\left(\alpha - 2\beta(1 - r(s_{1}))^{2}\right) + \mu(s_{2})\gamma_{2}\left(\alpha - 2\beta(1 - r(s_{2}))^{2}\right).$$

The first order condition yields the following optimality condition

$$\frac{\mu(s_1)}{\mu(s_2)} = \frac{r(s_1)}{r(s_2)} \cdot \frac{\gamma_2}{\gamma_1},\tag{7}$$

which in turns means that the wife will report her actual belief in response to the incentives provided by the quadratic scoring rule if and only if $\gamma_1 = \gamma_2$, or equivalently if and only if (u, μ) is a SI-SEU representation.

The previous example suggests that the entire applied literature that uses quadratic scoring rules to elicit beliefs relies on the implicit assumption that the subject (viz., the agent in our case) does not have any stakes in the underlying state space. This is because submitting a report corresponds to choosing an act from the menu

$$\{\sigma^r | r \in \Delta(S)\} \subseteq \mathcal{F}_S. \tag{8}$$

And, as we have already discussed extensively, such choice data cannot help us identify the agent's actual belief unless we assume state-independent utilities.

The need to assume state-independent utilities is not exclusive to cases where we use the quadratic scoring rule. This is because almost all elicitation mechanisms can be rewritten as menus of acts, like in (8). This is for instance the case for other scoring rules with deterministic rewards, binarized scoring rules, Karni mechanisms, matching probabilities, just to mention the most common ones.⁹ In the end, all these mechanisms yield traditional choice data, from which we cannot identify the agent's actual belief without running into the identification problem.¹⁰

Remark 2. Within the mechanism design literature, utility from money is typically linearly separable from the utility component that depends on the opponents' type. As a result, we can use standard belief elicitation tools to elicit one's beliefs about everyone else's type, despite being in a setting with interdependent types (Johnson et al., 1990; Jehiel et al., 2012). This is possible for the same

⁸In the scoring rule literature, an incentive-compatible (resp., strictly incentive-compatible) mechanism is called proper (resp., strictly proper).

⁹Examples of scoring rules with deterministic rewards include the logarithmic scoring rule and the spherical scoring rule (Good, 1971). There is a general characterization of this entire class by means of strictly convex indirect expected payoff functions (Savage, 1971). Binarized scoring rules pay in probability units to win a fixed prize in order to relax the risk-neutrality assumption (Hossain and Okui, 2013). Binarized scoring rules belong to the class of stochastic mechanisms, which also includes the Karni mechanisms (Karni, 2009; Tsakas, 2019) and the method of matching probabilities (Ducharme and Donnell, 1973; Baillon et al., 2018). For overviews of this literature, see Schlag and van der Weele (2013) and Schotter and Trevino (2014).

¹⁰The only exception is a handful of mechanisms which utilitize non-traditional choice data (Karni, 1999; Jaffray and Karni, 1999). Nevertheless, these mechanisms are quite cumbersome and data-demanding.

reason we can uniquely identify beliefs if we assume that utilities are quasi state-independent, as pointed out in Footnote 6. And vice versa, the same way we cannot pin down beliefs without an exogenous assumption —like quasi state-independent utilities— we will not be able to elicit beliefs about the opponents' types in mechanism design without the separability property which is implied by quasi-linearity.¹¹ \triangleleft

While the traditional mechanisms cannot help us with directly eliciting beliefs about the original state space S, they can still be used for eliciting the conditional beliefs about a proxy T given the original state space S. In practice, the agent will be asked to submit K reports r_1, \ldots, r_K , with $r_k \in \Delta(T)$ being her reported conditional belief about T given s_k . She will only be paid for her report r_k based on an incentive-compatible mechanism

$$\sigma_k: \Delta(T) \to \mathcal{F}_T,$$

if state s_k is eventually realized, e.g., if σ_k is a quadratic scoring rule, the agent will receive

$$\sigma_k^{r_k}(t) = \alpha - \beta \|r_k - \mathbf{1}_t\|^2$$

Note that in the end, the agent will only be paid for one of these mechanisms, as only one s_k will be eventually realized, meaning that this is essentially an implementation of the strategy method.¹²

5. Actual utility function

In this section, I will show how my main identification result allows me to obtain a well-founded definition of an *actual (state-dependent) utility function*. This is a very useful object, as it provides meaning to statements like "a dollar is twice as valuable at s_1 compared to s_2 " or "the utility of state s is equal w(s)", which are in turn commonly used in many applied settings, e.g., in the literature on motivated beliefs. Furthermore, actual utilities will be relevant for a recent literature that focuses on estimating tastes using directly elicited beliefs.

Note that these results will not be exclusive to my setting, and will also apply to other SEU models that uniquely identify actual beliefs. However, surprisingly, the point is largely ignored so far in the literature. I speculate that this is because most of the existing literature does not make the explicit distinction between "actual beliefs" and "beliefs that are uniquely identified within a SEU model". This distinction is necessary in order to identify "actual utilities" and is further discussed in Section 6.

Back in Example 1, recall that every SEU representation $(\tilde{u}, \tilde{\mu})$ of the wife's preferences satisfies

$$\frac{\tilde{\mu}(s_2)}{\tilde{\mu}(s_1)} = \frac{\tilde{\gamma}_1}{9\tilde{\gamma}_2}$$

where $\tilde{u}_1(q) = \tilde{\gamma}_1 q$ and $\tilde{u}_2(q) = \tilde{\gamma}_2 q$. Therefore, there is a duality between identification of beliefs and identification of relative marginal utilities. This exact duality is what the literature traditionally leverages to go from state-independent utilities to a unique belief, viz., by assuming $\gamma_1 = \gamma_2$ one can uniquely identify $\bar{\mu}(s_1) = 90\%$. In this paper, on the other hand, we will use this duality the other way around, viz., once we have identified the actual belief (using Theorem 1), we can uniquely identify a set of parameters (γ_1, γ_2) that will constitute the "actual utility function". For instance,

 $^{^{11}}$ Once again, I would like to thank the same referee that made the comment about quasi state-independent utilities.

¹²The fact that there is only one payment guarantees that there are no hedging opportunities (Blanco et al., 2010) which is a common concern in cases where the agent faces multiple decisions over acts.

in our running example, the actual belief is $\mu(s_1) = 0.10$ (Example 4, continued), and therefore the actual utility function can be written as

$$u_1(q) = \underbrace{\$1\beta}_{\gamma_1} q \text{ and } u_2(q) = \underbrace{\beta}_{\gamma_2} q,$$
(9)

where $\beta > 0$. The following theorem generalizes this result beyond my simple example, even in cases where there is no SI-SEU representation.

Theorem 2. Let $(\tilde{u}, \tilde{\mu})$ be an arbitrary SEU representation. Moreover, assume that the agent's actual belief $\mu \in \Delta(S)$ has been identified (via Theorem 1). Then, the class of utility functions $u: S \times Q \to \mathbb{R}$ for which (u, μ) is a SEU representation is characterized by

$$u_s = \alpha_s + \beta \frac{\tilde{\mu}(s)}{\mu(s)} \tilde{u}_s,\tag{10}$$

where $\alpha_s \in \mathbb{R}$ and $\beta > 0$, for all $s \in S$.

It should be stressed that throughout the paper the primitives are the preferences and the actual beliefs, while the actual utilities are still derived within the context of a SEU model. Using the actual utility function is only needed in order for us (i.e., the analyst) to correctly associate the agent's choices among acts with her actual belief.

Not surprisingly, as this is a Bayesian framework, the arrival of new information affects the actual beliefs, but not the actual tastes. So, the set of actual utility functions remains invariant when the agent updates her beliefs upon observing new information about S.

Remark 3. Interestingly, if the preferences admit a SI-SEU representation $(\bar{u}, \bar{\mu})$, Theorem 2 implies that the SEU function becomes

$$\mathbb{E}_{\mu}(u(f)) = \sum_{s \in S} \mu(s)w(s)\bar{u}(f_s), \tag{11}$$

where $w(s) = \bar{\mu}(s)/\mu(s)$ is the utility from state s. So, although state-independent preferences do not identify the actual belief, they still allow us to separately identify (state-independent) utility from a consequence at each state and utility from the state itself.

6. Decision-theoretic foundations

In this section, I provide axiomatic foundations for the conditions that appear in Definition 1, in order to clarify which assumptions are exogenous and which are testable with traditional choice data.

Aligned with Anscombe and Aumann (1963), let $Q = \Delta(X)$ be the set of lotteries over a finite set of prizes X.¹³ Take the extended state space $S \times T$, and denote the set of all acts by $\mathcal{F} := Q^{S \times T}$, with $f_{s,t} := f(s,t)$ being the lottery attached to state (s,t) by act f. The set of S-measurable acts is denoted by \mathcal{F}_S .¹⁴ Compound acts are defined in the usual way, i.e., $\lambda f + (1 - \lambda)g$ induces the outcome $\lambda f_{s,t} + (1 - \lambda)g_{s,t}$ at state $(s,t) \in S \times T$. For any $f,g \in \mathcal{F}$ and any $A \subseteq S \times T$, define the act f_Ag by

$$(f_A g)(s,t) = \begin{cases} f(s,t) & \text{if } (s,t) \in A, \\ g(s,t) & \text{if } (s,t) \notin A, \end{cases}$$

¹³The analysis can be easily extended to other decision-theoretic frameworks, e.g., Savage (1954) or Wakker (1989).

¹⁴As usual, an act $f \in \mathcal{F}$ is S-measurable (resp., T-measurable) whenever f(s,t) = f(s,t') for all $s \in S$ and $t, t' \in T$ (resp., f(s,t) = f(s',t) for all $s, s' \in S$ and $t \in T$).

i.e., $f_A g$ coincides with f in A and with g everywhere else.

The agent has (weak) preferences \succeq over \mathcal{F} . As usual, \succ and \sim denote the asymmetric part (viz., strict preference) and the symmetric part (viz., indifference) respectively. Conditional preferences \succeq_A are defined in the usual way: $f \succeq_A g$ if and only if $f_A h \succeq g_A h$ for all $h \in \mathcal{F}$. Throughout the paper, for notational simplicity, we write $\succeq_s := \succeq_{\{s\} \times T}$ and $\succeq_{s,t} := \succeq_{\{s\} \times \{t\}}$.

Note that the preference relation \succeq that we considered in Section 2 is not necessarily the same as \succeq restricted in \mathcal{F}_S , but rather as the conditional preference relation $\succeq_{S\times E}$ restricted in \mathcal{F}_S . To see that this the case, recall Example 4, where $\succeq \neq \boxdot$ and $\succeq_{t_2} = \trianglerighteq$, i.e., the wife's original preferences over \mathcal{F}_S are the same as the ones she would have had conditional on the husband having taken the placebo, but not unconditionally.

The latter also sheds some light on the relationship between this work and the distinction between small and large worlds of Savage (1954). On the one hand, similarly to Savage, introducing a proxy allows us to consider more contingencies, in that we also look at preferences over $(S \times T)$ -measurable acts, rather just S-measurable acts. That is, in the words of Savage (1954), we go from the smaller world S to the larger world $S \times T$. At the same time, in the small world we implicitly condition with respect to the event E. In this sense, unlike in Savage, when we go from the small to the large world, we are not only enriching the language (by considering more relevant aspects), but we are also adding more states that were previously ruled out from our consideration.

An event $A \subseteq S \times T$ is called null if for any two acts $f, g \in \mathcal{F}$ we have $f \sim_A g$. For the time being, to simplify presentation, we assume that there are no null states:¹⁵

 (A_0) FULL SUPPORT — No state $(s, t) \in S \times T$ is null.

Consider the following standard axioms:

(A₁) COMPLETENESS — For all $f, g \in \mathcal{F} : f \succeq g$ or $g \succeq f$.

(A₂) TRANSITIVITY — For all $f, g, h \in \mathcal{F}$: if $f \succeq g$ and $g \succeq h$, then $f \succeq h$.

- (A₃) CONTINUITY For all $f, g \in \mathcal{F} : \{g \in \mathcal{F} : f \succeq g\}$ and $\{g \in \mathcal{F} : g \succeq f\}$ are closed in \mathcal{F} .
- (A₄) INDEPENDENCE For all $f, g, h \in \mathcal{F}$, and for all $\lambda \in (0, 1) : f \succeq g$ if and only if $\lambda f + (1-\lambda)h \succeq \lambda f + (1-\lambda)h$.

These axioms form the basic premise for connecting beliefs to choices through some SEU representation (Karni et al., 1983). Formally, $(A_1) - (A_4)$ are satisfied if and only if a linear (state-dependent) utility function $u : S \times T \times Q \to \mathbb{R}$ together with actual joint belief $\pi \in \Delta(S \times T)$, represent the preferences \succeq via the SEU

$$\mathbb{E}_{\pi}\big(u(f)\big) = \sum_{s \in S} \sum_{t \in T} \pi(s, t) u_{s,t}(f_{s,t}).$$
(12)

That is, for all $f, g \in \mathcal{F}$,

$$f \succeq g \Leftrightarrow \mathbb{E}_{\pi}(u(f)) \ge \mathbb{E}_{\pi}(u(g)).$$
 (13)

Linearity of u implies that there is some (state-dependent) vNM utility function $v: S \times T \times X \to \mathbb{R}$, such that for every $(s,t) \in S \times T$ and every $q \in \Delta(X)$, the utility over consequences in (12) can be

¹⁵Without exogenously assuming that there are no null states, the definition of a proxy needs to be slightly adjusted. First of all, observe that the support of π can be uniquely identified from \succeq . Then, once we have pinned down the set of non-null states, conditions (P_0) , (P_1) and (P_3) will be respectively replaced by the following assumptions: (1) the agent has no stakes in T conditional on any non-null $s \in S$, (2) the uninformative event $E \subseteq T$ is non-null, and (3) the vectors in $\{\pi_T(\cdot|s)|s \text{ is non-null}\}$ are linearly independent.

rewritten as

$$u_{s,t}(q) = \sum_{x \in X} q(x) v_{s,t}(x).$$

For the rest of the paper we will work with u, keeping in mind that it is linear.

If we add (A_0) on top of $(A_1)-(A_4)$, the belief π becomes full-support. Of course, the identification problem still holds, i.e., for any $\tilde{\pi} \in \Delta(S \times T)$ there exists some linear \tilde{u} such that $\mathbb{E}_{\pi}(u(f)) = \mathbb{E}_{\tilde{\pi}}(\tilde{u}(f))$ for all $f \in \mathcal{F}$, meaning that $(\tilde{u}, \tilde{\pi})$ also represents \succeq . I henceforth treat $(A_0) - (A_4)$ as prerequisite assumptions, not because they cannot be tested, but rather because the belief identification problem is formally defined only once they hold.

Then, assuming that $(A_0) - (A_4)$ are satisfied, I will proceed to restrict the class of SEU representations that are consistent with T being a proxy for S, given that π_T^{obj} has been already labelled as the objective marginal belief. The latter means that there are commonly known objective facts from which π_T^{obj} is obtained.

Definition 2. A full-support SEU representation $(\bar{u}, \bar{\pi})$ is called *Conditionally State-Independent* (*abbrev., CSI-SEU*) whenever \bar{u} is *S*-measurable, i.e., whenever $\bar{u}_{s,t}(q) = \bar{u}_{s,t'}(q)$ for all $s \in S$, all $t, t' \in T$, and all $q \in Q$. Furthermore, a CSI-SEU representation $(\bar{u}, \bar{\pi})$ is called *Proxy-Consistent* (*abbrev., PC-SEU*), whenever (a) $\bar{\pi}_T = \pi_T^{\text{obj}}$, and (b) $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ are linearly independent.

Being able to represent \succeq with a PC-SEU $(\bar{u}, \bar{\pi})$ is the minimal necessary condition for T to be a proxy, viz., S-measurability of \bar{u} is necessary for (P_0) ; agreement of $\bar{\pi}_T$ with π_T^{obj} is necessary for (P_2) ; and linear independence of $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ is necessary for (P_3) . However, existence of a PC-SEU representation is not sufficient by any means, because $(P_0) - (P_3)$ involve exogenous assumptions. For instance, even if we establish that $(\bar{u}, \bar{\pi})$ is a PC-SEU representation, it is never possible to conclude that $\bar{\pi}$ is the actual joint belief π , for the same reasons we could not conclude $\bar{\mu} = \mu$. In this sense, we can only reject the hypothesis that T is a proxy by showing that \succeq does not have a PC-SEU representation, but we cannot verify it even if a PC-SEU representation exists. I further discuss this issue at the end of this section. For the time being, I will axiomatize the class of PC-SEU representations, beginning with the following axiom:

(A₅) LOCAL STATE-MONOTONICITY — For all $s \in S$, for all $t, t' \in T$, and for all $p, q \in Q : p \succeq_{s,t} q$ if and only if $p \succeq_{s,t'} q$.

This is simply a weakening of the standard state-monotonicity axiom (Anscombe and Aumann, 1963), in that it postulates state-independent preferences only across states in $\{s\} \times T$, rather than across all states in $S \times T$. It is not difficult to show that, together with $(A_0) - (A_4)$, it axiomatizes CSI-SEU. But most importantly, analogously to Anscombe and Aumann (1963), it guarantees that the conditional belief $\bar{\pi}_T(\cdot|s)$ is uniquely identified for every $s \in S$, i.e., all CSI-SEU representations yield the same $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ (Lemma A1).

In order to formally state my next two axioms, I first need to introduce some additional notation. Let $\mathcal{E} = \{f^{\text{high}}, f^{\text{low}}\}$ be a binary menu of strict-dominance-ordered S-measurable acts, i.e., for all states (s, t), it is the case that $f^{\text{high}}_{s,t} =: f^{\text{high}}_s$ and $f^{\text{low}}_{s,t} =: f^{\text{low}}_s$, and moreover

$$f^{\text{high}} \succ_{s,t} f^{\text{low}}$$

For instance, if \succeq satisfies (A_5) , one such menu is formed if we take f^{high} and f^{low} to be a \succeq -maximal and a \succeq -minimal act respectively. If such menu exists, consider the lattice (with respect to the dominance relation),

$$\mathcal{F}_{\mathcal{E}} := \{ f \in \mathcal{F} : f^{\text{high}} \succeq_{s,t} f \succeq_{s,t} f^{\text{low}} \text{ for all } (s,t) \in S \times T \}.$$
(14)

Unlike \mathcal{E} itself, $\mathcal{F}_{\mathcal{E}}$ does not necessarily contain only *S*-measurable acts. For instance, in our earlier example (where f^{high} and f^{low} are \succeq -maximal and \succeq -minimal acts respectively), $\mathcal{F}_{\mathcal{E}}$ is the entire set \mathcal{F} . Since at least one of the two relations in (14) is strict, by continuity, for every $f \in \mathcal{F}_{\mathcal{E}}$ and every state (s, t) there exists a unique $\lambda_{s,t}^{\mathcal{E},f} \in [0, 1]$ such that

$$f \sim_{s,t} (1 - \lambda_{s,t}^{\mathcal{E},f}) f^{\text{low}} + \lambda_{s,t}^{\mathcal{E},f} f^{\text{high}}.$$
 (15)

Thus, each $f \in \mathcal{F}_{\mathcal{E}}$ is associated with a unique vector $\lambda^{\mathcal{E},f} \in [0,1]^{S \times T}$. Then, define

$$\mathcal{T}_{\mathcal{E}} = \{ f \in \mathcal{F}_{\mathcal{E}} : \lambda^{\mathcal{E}, f} \text{ is } T \text{-measurable} \}.$$
(16)

In general, T-measurability of $\lambda^{\mathcal{E},f}$ neither implies nor is it implied by T-measurability of f.

Finally, for every $f \in \mathcal{F}$, define the S-measurable act

$$f_s^{\text{obj}} := \sum_{t \in T} \pi_T^{\text{obj}}(t) f_{s,t}.$$
(17)

The idea is that f^{obj} averages across the lotteries that f assigns to the different states in $\{s\} \times T$ with respect to the exogenously given objective marginal π_T^{obj} , and subsequently assigns to each state (s,t) the average lottery f_s^{obj} .

Then, I am ready to introduce the two new axioms:

- (A₆) OBJECTIVE MIXTURE INDIFFERENCE There is some binary strict-dominance-ordered $\mathcal{E} \subseteq \mathcal{F}_S$ such that for all $f \in \mathcal{T}_{\mathcal{E}} : f^{\text{obj}} \sim f$.
- (A₇) UNIQUE EXTENSION For any binary strict-dominance-ordered $\mathcal{E} \subseteq \mathcal{F}_S$, and for any \succeq' satisfying $(A_0) (A_5)$: if $\succeq = \succeq'$ in $\mathcal{T}_{\mathcal{E}}$ then $\succeq = \succeq'$ in $\mathcal{F}_{\mathcal{E}}$.

Axiom (A_6) will guarantee that, there is a CSI-SEU representation $(\bar{u}, \bar{\pi})$ such that $\bar{\pi}_T = \pi_T^{\text{obj}}$ (Lemma A2), i.e., it essentially allows us to calibrate against the objective marginal π_T^{obj} , similarly in spirit to the calibration exercise of Anscombe and Aumann (1963). This axiom is of somewhat nonstandard nature as it is defined with respect to the exogenously given π_T^{obj} . It essentially postulates that if we replace every conditional belief $\bar{\pi}_T(\cdot|s)$ with the objective belief π_T^{obj} , the implied preferences over $\mathcal{T}_{\mathcal{E}}$ will not change.

Axiom (A_7) will guarantee that the uniquely identified beliefs $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ that have been previously obtained as part of every CSI-SEU representation are linearly independent (Lemma A3).

Theorem 3. There exists a PC-SEU representation $(\bar{u}, \bar{\pi})$ if and only if \succeq satisfies $(A_0) - (A_7)$. Furthermore, $\bar{\pi}$ is uniquely identified from \succeq .

The first important implication of the previous result is that it allows us to formally state the exogenous assumptions in the definition of a proxy. These assumptions are: (1) the objective marginal belief π_T^{obj} agrees with the actual marginal belief π_T , (2) the conditional beliefs $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ agree with the actual conditional beliefs $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$, and (3) the conditional belief $\bar{\pi}_S(\cdot|E)$ agrees with the actual belief μ . Not surprisingly, all three of them have to do with the relationship between $\bar{\pi}$ and the actual beliefs.

The second important implication of Theorem 3 goes back to the long-standing debate on the definition of subjective probability (e.g., Grant and van Zandt, 2008; Karni, 2014, and references therein). Although I personally subscribe to the more modern approach (according to which actual

beliefs exist), this paper can be easily rewritten along the lines of the classical view (according to which the only observable primitive is the preference relation, and therefore beliefs should be simply defined as the probability measure that we uniquely identify within a class of SEU representations). Indeed, using Theorem 3, I could have simply labelled the uniquely identified $\bar{\pi}_S(\cdot|E)$ as the agent's belief, without going into the whole discussion on whether this is her actual beliefs, or even on whether an actual belief exists in the first place.

7. Belief identification in practice

7.1. Availability of proxies

Suitable proxies can be classified into two broad categories, depending on whether we rely on existing observational data or on experimental data that we specifically collect for the purpose of our elicitation exercise.

Starting with the former, sometimes a suitable proxy appears spontaneously, like for instance in Example 4, where a clinical trial incidentally happens to take place. Other times a suitable proxy is picked from publicly available statistics, like for instance in Example 5, where the percentage of people who carry the specific gene is known. Such proxies will often fit our definition well, but it is unclear whether they will always be readily available.

In this last case we turn to the second category, where we design an experiment —prior to the elicitation task— with the specific aim of constructing a suitable proxy. Fortunately for us, it is essentially always feasible to construct such a proxy. To do this, we first generate an informative signal, then we distort it, and finally we elicit the agent's conditional probabilities of the signal having been distorted given each realization of the original state space, as I have already briefly described in Example 6. Let me elaborate a bit further in order to showcase the easiness of constructing such proxies.

Begin with collecting (conditionally) independent predictions about the husband's health outcome from a group of experts. Examples of such experts include doctors, medical students, AI tools, etc. The wife knows which experts are in the group, but does not know the prediction of each individual expert. From her point of view, the informative signal is given by her perceived reliability of a randomly drawn expert, i.e., by the subjective probability that she attaches to a randomly picked expert correctly predicting her husband's recovery, as well as the subjective probability that she assigns to a randomly picked expert correctly predicting that her husband remains paralyzed (Figure 2). Importantly, we remain agnostic on what the wife thinks about what these error probabilities



Figure 2: The wife's perceived reliability of a randomly drawn expert are described by the two error probabilities, i.e., ε and δ , which correspond her subjective estimate of a random expert's type 1 and type 2 error.

could look like. But at the end of the day this is not a problem for our purposes, as we only care about the wife believing that the signal contains some information, i.e., our only requirement is $1 - \varepsilon \neq \delta$. This means, for instance, that we do not require ε and δ to coincide with an expert's actual reliability.

Now, we distort the signal by mixing the prediction of a randomly drawn expert with noise. Here is one of the many potential examples on how we could implement such distortion in practice. Suppose that inside an urn we place a green ball for each expert predicting "recovers" and a red ball for each expert predicting "paralyzed". All these balls will be henceforth called informative, as each of them corresponds to an actual prediction. Then, we add into the same urn some more balls, half of them being green and half being red. We call these balls uninformative, as they do not correspond to anyone's prediction. Then, we define the set

 $T = \{ \text{informative ball } (t_1), \text{ uninformative ball } (t_2) \}.$

The wife's joint belief over states and signal realizations, given an informative ball (resp., given an uninformative ball), is the distribution on the left hand-side table (resp., right hand-side table):

| | recovers (s_1) | paralyzed (s_2) | | recovers (s_1) | paralyzed (s_2) |
|--------------------------|---------------------------|----------------------|------------|-------------------|-------------------|
| green ball | $(1-\varepsilon)\mu(s_1)$ | $\delta\mu(s_2)$ | green ball | $0.5\mu(s_1)$ | $0.5\mu(s_2)$ |
| red ball | $arepsilon \mu(s_1)$ | $(1-\delta)\mu(s_2)$ | red ball | $0.5\mu(s_1)$ | $0.5\mu(s_2)$ |
| informative ball (t_1) | | | uninformat | tive ball (t_2) | |

This follows directly from Figure 2 above.

Suppose that a ball is randomly drawn from the urn. For illustration purposes, let this ball be green. The wife is told that the ball is green. She is also told the proportion of informative balls among the green balls, which is henceforth denoted by $\pi_T^g(t_1)$. Then, her joint belief over the product space $S \times T$ will be given by the following table:

$$\begin{array}{c} \text{recovers } (s_1) & \text{paralyzed } (s_2) \end{array}$$

$$\begin{array}{c} \text{informative ball } (t_1) \end{array} & \left[\begin{array}{c} \pi_T^g(t_1) \frac{(1-\varepsilon)\mu(s_1)}{(1-\varepsilon)\mu(s_1)+\delta\mu(s_2)} & \pi_T^g(t_1) \frac{\delta\mu(s_2)}{(1-\varepsilon)\mu(s_1)+\delta\mu(s_2)} \\ \text{uninformative ball } (t_2) \end{array} \right] \\ \pi_T^g(t_2)\mu(s_1) & \pi_T^g(t_2)\mu(s_2) \end{array}$$

Note that all the assumptions of Definition 1 are consistent with this joint belief, as long as $1 - \varepsilon \neq \delta$, meaning that T can be used as a suitable proxy. In particular, as required by (P_1) , the event $\{t_2\}$ is uninformative. Moreover, in line with (P_2) , the marginal distribution $\pi_T^g \in \Delta(T)$ is commonly known. Finally, as postulated by (P_3) , the proxy T is correlated with S. Thus, given that it is also reasonable to also assume (P_0) (see Example 6), we are comfortable with using T as a proxy.

Concluding, it is exactly the flexibility in picking any information sources for generating the signal, combined with the fact that we do not impose any assumption on what the wife believes about the source's reliability, what makes this approach essentially always feasible.

7.2. Applications revisited

As pointed out in the introduction, there are several applications where state-dependent preferences can potentially lead to measurement error when eliciting beliefs. Let me illustrate here how my proxy method could be useful in mitigating this problem in a couple of those application.

Suppose that we want to elicit the beliefs of partisan voters about the outcome of the upcoming presidential election. It is well known that subjects systematically differ in their stated beliefs depending on which candidate they support. The question is whether such differences should be fully attributed to overconfidence —as the literature on politically motivated beliefs suggests— or whether part of these differences should be attributed to measurement error because of state-dependent preferences. If such measurement error is positive, we can conclude that the standard methodology overestimates the effect of motivation. But before explaining how my method can help us solve the problem, let me explain why the identification problem is likely to arise here. The preference that partisan voters exhibit for their own candidate winning can be associated with expectation of some personal gain in case this happens, e.g., otherwise they may choose to relocate abroad in which case their purchasing power will drop, or they may suffer psychologically and consumption is a way to compensate. In either case, it is likely that their utility from money is state-dependent.

Having said this, within this application, we can always use a variant of the experimental design we presented in Section 7.1 above, i.e., we can always present the voter with a forecast or a poll, and then elicit her conditional beliefs about this forecast/poll coming from an actual source conditional on her preferred candidate winning, as well as conditional on the opponent winning.

Now, let me turn to a second application from the ones I presented in the introduction. Suppose that we want to use the forecast of a financial analyst as input for our decision on whether to buy a certain stock or not. However, although we have calibrated this analyst's utility for money, we cannot observe her existing portfolio and therefore we do not know whether she has already invested in this stock. Thus, the analyst's state-dependent preferences enter the picture in the form of unobservable side payments, like in Example 3.

Within this application, we can again use an experimental design along the lines of what I presented in Section 7.1. In particular, suppose that I present the analyst with the forecast of another expert, and subsequently elicit her conditional beliefs of this other forecast coming from an actual expert conditional on the stock's price going up, as well as conditional on the stock's price going down.

8. Conclusion

In this paper, I proposed a novel approach for identifying subjective beliefs without exogenously imposing the awkward assumption of state-independent utilities. The idea is to enlarge the state space by introducing a second dimension, which I call a proxy. The key property of a proxy is that the agent has no stakes in its realization conditional on the original state space. It is exactly this property that allows us to uniquely identify the agent's conditional beliefs about the proxy given each realization of the original state space, using a variant of the strategy method. This method is less data-demanding and more flexible compared to other methods in the literature. Thus, it is not just theoretically sound, but also tractable. Of course, there are various important details which pertain to the experimental implementation of the methodology and need to be taken into account in practice, e.g., calibrating for updating biases (Benjamin, 2019), accounting for the possibility of uncertain beliefs (Enke and Graeber, 2023), dealing with the complexity that comes from explicitly spelling out the incentives of the elicitation mechanism (Danz et al., 2022) or from the need to

perform contingent reasoning (Esponda and Vespa, 2014, 2023), etc. Nevertheless, addressing all these issues is outside the scope of the present paper, as they are all orthogonal to the fundamental long-standing belief identification problem that this paper tackles.

A. Proofs

Proof of Theorem 1. By (P_0) all conditional beliefs $\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)$ are uniquely identified by any standard incentive-compatible elicitation task (e.g., a binarized scoring rule) applied K times, once for each $s \in S$. Since π_T is in the convex hull of $\{\pi_T(\cdot|s_1), \ldots, \pi_T(\cdot|s_K)\}$, there is $\lambda = (\lambda_1, \ldots, \lambda_K) \in \mathbb{R}^K_+$ with $\lambda_1 + \cdots + \lambda_K = 1$ such that

$$\pi_T = \sum_{k=1}^K \lambda_k \pi_T(\cdot | s_k), \tag{A.1}$$

where, by (P_2) , we know π_T . By the law of total probability, π_S solves this system. Moreover, it is the unique solution if and only if matrix

$$\Pi = \begin{bmatrix} \pi_T(t_1|s_1) & \cdots & \pi_T(t_N|s_1) \\ \vdots & \ddots & \vdots \\ \pi_T(t_1|s_K) & \cdots & \pi_T(t_N|s_K) \end{bmatrix}$$

is such that rank(Π) = K. The latter holds if and only if (P_3) is satisfied. Then, by the chain rule of probability, we obtain $\pi(s,t) = \pi_S(s)\pi_T(t|s)$. Finally, by the definition of conditional probability, we have

$$\pi_S(s|E) = \frac{\pi(\{s\} \times E)}{\pi_T(E)}.$$
(A.2)

And, by (P_1) , we obtain $\mu = \pi_S(\cdot | E)$, which completes the proof.

Proof of Theorem 2. First, we will prove that, if u is defined as in (10) then (u, μ) is a SEU representation. Take arbitrary $(\alpha_1, \ldots, \alpha_K) \in \mathbb{R}^K$ such that $\sum_{k=1}^K \mu(s_k)\alpha_k = \alpha$, and $\beta > 0$. Then, for all $f \in \mathcal{F}_S$,

$$\mathbb{E}_{\mu}(u(f)) = \alpha + \beta \mathbb{E}_{\tilde{\mu}}(\tilde{u}(f)), \tag{A.3}$$

which completes this part of the proof.

Let us now prove the converse, i.e., if (\tilde{u}, μ) is a SEU then \tilde{u} is necessarily defined as in (10). For each $k = 1, \ldots, K$, define the (convex) range of the original state-utility function, $Y_k := u_k(Q) \subseteq \mathbb{R}$. Then, it is obvious that, the state-utility function \tilde{u}_k is obtained via a strictly increasing transformation of u_k , i.e., there is a continuous strictly increasing $\phi_k : Y_k \to \mathbb{R}$ such that

$$\tilde{u}_k = \phi_k \circ u_k. \tag{A.4}$$

This is because both u_k and \tilde{u}_k represent the same state-preferences.

We will now show that ϕ_k is linear. Take an arbitrary $y \in int(Y_1 \times \cdots \times Y_K)$, and for each $k = 1, \ldots, K$ define $y^k \in \mathbb{R}^K$ by

$$y_{\ell}^{k} = \begin{cases} y_{\ell} & \text{if } \ell \neq k, \\ y_{k} + \delta/\mu(s_{k}) & \text{if } \ell = k. \end{cases}$$
(A.5)

Since y is an interior point, there exists some $\delta_y > 0$ such that $y^k \in Y_1 \times \cdots \times Y_K$ for all $\delta \in (0, \delta_y)$, and every $k = 1, \ldots, K$. Hence, for each $k = 1, \ldots, K$, there exists an act $f^k \in \mathcal{F}_S$ such that, for every $\ell = 1, \ldots, K$, it is the case that

$$u_{\ell}(f_{\ell}^k) = y_{\ell}^k, \tag{A.6}$$

| _ | |
|---|--|
| | |
| | |
| | |
| | |

Therefore, by construction, we obtain

$$\mathbb{E}_{\mu}(u(f^{k})) = \sum_{\ell=1}^{K} \mu(s_{\ell}) y_{\ell}^{k} = \sum_{\ell=1}^{K} \mu(s_{\ell}) y_{\ell} + \delta.$$
(A.7)

Since the right hand-side does not depend on k, we have $f^1 \sim \cdots \sim f^K$. Then, since (\tilde{u}, μ) is a SEU representation, we obtain

$$\mathbb{E}_{\mu}(\tilde{u}(f^1) = \dots = \mathbb{E}_{\mu}(\tilde{u}(f^K).$$
(A.8)

By combining (A.4) and (A.6), we get

$$\mathbb{E}_{\mu}(\tilde{u}(f^k) = \sum_{\ell=1}^{K} \mu(s_\ell) \phi_\ell(y_\ell^k), \tag{A.9}$$

and consequently, by (A.8), it follows that for any two distinct $k, \ell = 1, \ldots, K$,

$$\sum_{m=1}^{K} \mu(s_m) \phi_k(y_m^k) = \sum_{m=1}^{K} \mu(s_m) \phi_\ell(y_m^\ell).$$

Using the definition of y^k and y^ℓ from (A.5), the previous equation can be rewritten as

$$\mu(s_k)\phi_k(y_k + \delta/\mu(s_k)) + \mu(s_\ell)\phi_\ell(y_\ell) = \mu(s_k)\phi_k(y_k) + \mu(s_\ell)\phi_\ell(y_\ell + \delta/\mu(s_\ell)).$$

Rearranging terms, dividing both sides with δ , and taking right limits, yields

$$\phi_{k+}'(y_k) = \lim_{\delta \downarrow 0} \frac{\phi_k(y_k + \delta/\mu(s_k)) - \phi_k(y_k)}{\delta/\mu(s_k)} = \lim_{\delta \downarrow 0} \frac{\phi_\ell(y_\ell + \delta/\mu(s_\ell)) - \phi_\ell(y_\ell)}{\delta/\mu(s_\ell)} = \phi_{\ell+}'(y_\ell).$$
(A.10)

Note that the right derivatives are well-defined as the respective domains Y_k and Y_ℓ are convex sets. We repeat this exercise with any $y' \in int(Y_1 \times \cdots \times Y_K)$ which agrees with y at all coordinates except k, i.e., $y_k \neq y'_k$ and $y_\ell = y'_\ell$ for all $\ell \neq k$. Thus, we obtain

$$\phi'_{k+}(y'_k) = \phi'_{\ell+}(y'_\ell). \tag{A.11}$$

But, since $y_{\ell} = y'_{\ell}$, it follows directly that, for any two $y_k, y'_k \in int(Y_k)$,

$$\phi'_{k+}(y_k) = \phi'_{k+}(y'_k), \tag{A.12}$$

i.e., the right derivative of ϕ_k is constant in the interior of its domain. Therefore, together with continuity (including at the boundaries in case those belong to Y_k) it implies that ϕ_k is linear for all $k = 1, \ldots, K$. But then, by (A.10), the slope of ϕ_k is the same for all $k = 1, \ldots, K$, which completes the proof.

Lemma A1. There is a CSI-SEU representation $(\bar{u}, \bar{\pi})$ if and only if \succeq satisfies $(A_0) - (A_5)$. Moreover, the conditional beliefs $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ are uniquely identified from \succeq .

Proof. It follows directly from applying Anscombe and Aumann (1963) to \succeq_s for every $s \in S$.

Lemma A2. Suppose that \succeq satisfies $(A_0) - (A_5)$, and let $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ be the uniquely identified conditional beliefs from Lemma A1. Then, there is some CSI-SEU representation $(\bar{u}, \bar{\pi})$ with $\bar{\pi}_T = \pi_T^{\text{obj}}$ if and only if \succeq satisfies (A_6) .

Proof. NECESSITY: By (A_0) and (A_3) , there exists some $\delta > 0$, and a pair of lotteries $q_s^{\text{high}}, q_s^{\text{low}} \in Q$ for each $s \in S$, such that

$$\bar{u}_s(q_s^{\text{high}}) - \bar{u}_s(q_s^{\text{low}}) = \delta$$

Then, take $\mathcal{E} = \{f^{\text{high}}, f^{\text{low}}\}$ such that, for every $s \in S$ and $t \in T$,

$$\begin{array}{rcl}
f_{s,t}^{\text{high}} & := & q_s^{\text{high}}, \\
f_{s,t}^{\text{low}} & := & q_s^{\text{low}}.
\end{array}$$

Note that, for every $f \in \mathcal{F}_{\mathcal{E}}$,

$$\bar{u}_s(f_{s,t}) = \lambda_{s,t}^{\mathcal{E},f} \delta + \bar{u}_s(q_s^{\text{low}}) \tag{A.13}$$

Hence, for all $f \in \mathcal{T}_{\mathcal{E}}$, by *T*-measurability of $\lambda^{\mathcal{E},f}$,

$$\mathbb{E}_{\bar{\pi}}(\bar{u}(f)) = \delta \sum_{t \in T} \bar{\pi}_T(t) \lambda_t^{\mathcal{E}, f} + \sum_{s \in S} \bar{\pi}_S(s) \bar{u}_s(q_s^{\text{low}}).$$
(A.14)

Furthermore, for the same $f \in \mathcal{T}_{\mathcal{E}}$, we obtain

$$\mathbb{E}_{\bar{\pi}}(\bar{u}(f^{\text{obj}})) = \sum_{s \in S} \bar{\pi}_S(s)\bar{u}_s(f_s^{\text{obj}})$$
(A.15)

$$= \sum_{s \in S} \bar{\pi}_S(s) \sum_{t \in T} \bar{\pi}_T(t) \bar{u}_s(f_{s,t})$$
(A.16)

$$= \delta \sum_{t \in T} \bar{\pi}_T(t) \lambda_t^{\mathcal{E}, f} + \sum_{s \in S} \bar{\pi}_S(s) \bar{u}_s(q_s^{\text{low}}).$$
(A.17)

where (A.15) follows from S-measurability of f^{obj} , (A.16) follows from the definition of f^{obj} combined with $\bar{\pi}_T = \pi_T^{obj}$, and (A.17) follows from (A.13) combined with T-measurability of $\lambda^{\mathcal{E},f}$. Hence, $f \sim f^{obj}$, i.e., (A₆) holds.

SUFFICIENCY: Let $\mathcal{E} = \{f^{\text{high}}, f^{\text{low}}\}\$ be the menu of S-measurable acts such that $f \sim f^{\text{obj}}$ for all $f \in \mathcal{F}_{\mathcal{E}}$. Take an arbitrary CSI-SEU $(\hat{v}, \hat{\pi})$ of \succeq . Note that there exist $\alpha_s \in \mathbb{R}$ and $\beta_s > 0$ such that $\alpha_s + \beta_s \hat{u}_s(f_s^{\text{high}}) = 1$ and $\alpha_s + \beta_s \hat{u}_s(f_s^{\text{low}}) = 0$. Then, define the normalized S-measurable \bar{u} by

$$\bar{u}_s := \alpha_s + \beta_s \hat{u}_s \tag{A.18}$$

for each $s \in S$. Moreover, for every $s \in S$, define the new marginal belief

$$\bar{\pi}_{S}(s) := \frac{\hat{\pi}_{S}(s)/\beta_{s}}{\sum_{s' \in S} \hat{\pi}_{S}(s')/\beta_{s'}},\tag{A.19}$$

and observe that for every $f \in \mathcal{F}$, we obtain

$$\mathbb{E}_{\hat{\pi}}(\hat{u}(f)) = \alpha + \beta \mathbb{E}_{\bar{\pi}}(\bar{u}(f)), \qquad (A.20)$$

where $\alpha := \sum_{s \in S} \alpha_s \bar{\pi}_S(s)$ and $\beta := \sum_{s' \in S} \hat{\pi}_S(s') / \beta_{s'} > 0$ are both constants, and the joint belief $\bar{\pi}$ is defined by

$$\bar{\pi}(s,t) := \bar{\pi}_S(s)\hat{\pi}_T(t|s). \tag{A.21}$$

Hence, the pair $(\bar{v}, \bar{\pi})$ is a CSI-SEU.

Fix an arbitrary $t \in T$, and take the act

$$\langle t \rangle := f_{S \times \{t\}}^{\text{high}} f^{\text{low}},$$

meaning that for every state (s, t), the state-utility becomes

$$\bar{u}_s(\langle t \rangle_{s,t'}) = \begin{cases} 1 & \text{if } t = t' \\ 0 & \text{if } t \neq t' \end{cases}$$

Hence, by construction, we have

$$\mathbb{E}_{\bar{\pi}}(\bar{u}(\langle t \rangle)) = \sum_{s \in S} \bar{\pi}_S(s) \bar{\pi}_T(t|s) = \bar{\pi}_T(t).$$
(A.22)

Moreover, by the definition of $\langle t \rangle^{\rm obj}$, we obtain

$$\mathbb{E}_{\bar{\pi}}(\bar{u}(\langle t \rangle^{\mathrm{obj}})) = \sum_{s \in S} \bar{\pi}_S(s) \pi_T^{\mathrm{obj}}(t) = \pi_T^{\mathrm{obj}}(t).$$
(A.23)

Finally, note that by construction, the act $\langle t \rangle$ belongs to $\mathcal{T}_{\mathcal{E}}$. Hence, by (A_6) , we have $\bar{\pi}_T = \pi_T^{\text{obj}}$.

Lemma A3. Suppose that \succeq satisfies $(A_0) - (A_5)$, and let $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ be the uniquely identified conditional beliefs from Lemma A1. Then, $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ are linearly independent if and only if \succeq satisfies (A_7) .

Proof. INTERMEDIATE STEP: Fix an arbitrary menu of S-measurable acts $\mathcal{E} = \{f^{\text{high}}, f^{\text{low}}\}$ such that $f^{\text{high}} \succ_{s,t} f^{\text{low}}$ for all states (s,t). Take a CSI-SEU representation $(\hat{v}, \hat{\pi})$ of \succeq , and like in the proof of sufficiency in the previous lemma, obtain another CSI-SEU representation $(\bar{v}, \bar{\pi})$ such that $\bar{u}_s(f_s^{\text{high}}) = 1$ and $\bar{u}_s(f_s^{\text{low}}) = 1$. Hence, for every $f \in \mathcal{F}_{\mathcal{E}}$, and every state (s,t), we have

$$\bar{u}_s(f_{s,t}) = \lambda_{s,t}^{\mathcal{E},f}$$

Hence, for every $f \in \mathcal{T}_{\mathcal{E}}$, by *T*-measurability of $\lambda^{\mathcal{E},f}$, we get

$$\mathbb{E}_{\bar{\pi}}(\bar{u}(f)) = \sum_{t \in T} \lambda_t^{\mathcal{E}, f} \sum_{s \in S} \bar{\pi}_S(s) \bar{\pi}_T(t|s).$$
(A.24)

SUFFICIENCY: Suppose that $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ are not linearly independent. Hence, there exists some $\bar{\pi}'_S \in \Delta(S)$ with $\bar{\pi}'_S \neq \bar{\pi}_S$ such that

$$\sum_{s \in S} \bar{\pi}'_{S}(s)\bar{\pi}_{T}(\cdot|s) = \sum_{s \in S} \bar{\pi}_{S}(s)\bar{\pi}_{T}(\cdot|s) = \bar{\pi}_{T}.$$
(A.25)

Consider the preference relation \succeq' which is represented by $(\bar{v}, \bar{\pi}')$. By S-measurability of \bar{u} , it follows that \succeq' satisfies $(A_0) - (A_5)$. Moreover, by (A.24), it follows that $\mathbb{E}_{\bar{\pi}}(\bar{u}(f)) = \mathbb{E}_{\bar{\pi}'}(\bar{u}(f))$ for every $f \in \mathcal{T}_{\mathcal{E}}$, i.e.,

$$\succeq = \succeq' \text{ in } \mathcal{T}_{\mathcal{E}}. \tag{A.26}$$

For any $\lambda_0, \lambda_1, \lambda_2 \in (0, 1)$, define the following acts that belong to $\mathcal{F}_{\mathcal{E}}$:

$$f^{0} := \lambda_{0} f^{\text{high}} + (1 - \lambda_{0}) f^{\text{low}},$$

$$f^{1} := \lambda_{1} f^{\text{high}} + (1 - \lambda_{1}) f^{\text{low}},$$

$$f^{2} := \lambda_{2} f^{\text{high}} + (1 - \lambda_{2}) f^{\text{low}}.$$

Then, observe that the following equivalences hold for all $s \in S$:

$$f^{1}_{\{s\}\times T}f^{2} \succeq f^{0} \Leftrightarrow \bar{\pi}_{S}(s)\lambda_{1} + (1 - \bar{\pi}_{S}(s))\lambda_{2} \ge \lambda_{0},$$

$$f^{1}_{\{s\}\times T}f^{2} \succeq' f^{0} \Leftrightarrow \bar{\pi}'_{S}(s)\lambda_{1} + (1 - \bar{\pi}'_{S}(s))\lambda_{2} \ge \lambda_{0}.$$

By $\bar{\pi}_S \neq \bar{\pi}'_S$, there is some $s \in S$ such that $\bar{\pi}_S(s) > \bar{\pi}'_S(s)$. Hence, for some $0 < \lambda_2 < \lambda_0 < \lambda_1 < 1$, we obtain $f^1_{\{s\} \times T} f^2 \succeq f^0$ and $f^1_{\{s\} \times T} f^2 \not\succeq' f^0$, meaning that

$$\succeq \neq \succeq' \text{ in } \mathcal{F}_{\mathcal{E}}.\tag{A.27}$$

Hence, \succeq is not uniquely extended from $\mathcal{T}_{\mathcal{E}}$ to $\mathcal{F}_{\mathcal{E}}$, meaning that (A_6) is violated.

NECESSITY: Suppose that $\bar{\pi}_T(\cdot|s_1), \ldots, \bar{\pi}_T(\cdot|s_K)$ are linearly independent. Take an arbitrary \succeq' satisfying $(A_0) - (A_5)$, such that $\succeq = \succeq'$ in $\mathcal{T}_{\mathcal{E}}$. This implies that $\succeq_s = \succeq'_s$ for all $s \in S$, as $\mathcal{T}_{\mathcal{E}}$ places restrictions only across T. Therefore, the CSI-SEU representations $(\bar{u}, \bar{\pi})$ and $(\bar{u}', \bar{\pi}')$ that we constructed in the intermediate step, are such that $\bar{u} = \bar{u}'$ and $\bar{\pi}_T(\cdot|s) = \bar{\pi}'_T(\cdot|s)$ for all $s \in S$.

Suppose that $\bar{\pi}_S \neq \bar{\pi}'_S$. Then, by linear independence, there exists some $t \in T$ such that

$$\bar{\pi}_T'(t) = \sum_{s \in S} \bar{\pi}_S'(s) \bar{\pi}_T(t|s) < \sum_{s \in S} \bar{\pi}_S(s) \bar{\pi}_T(t|s) = \bar{\pi}_T(t).$$
(A.28)

Pick some $\lambda \in (0,1)$ such that $\bar{\pi}'_T(t) < \lambda < \bar{\pi}_T(t)$, and define the act h such that $\bar{u}_{s,t}(h_{s,t}) = \lambda$ for every (s,t). It is not difficult to verify that such act exists and $h \in \mathcal{T}_{\mathcal{E}}$. Moreover, we have $f_{S \times \{t\}}^{\text{high}} f^{\text{low}} \in \mathcal{T}_{\mathcal{E}}$. Hence, by (A.24), we obtain

$$f_{S \times \{t\}}^{\text{high}} f^{\text{low}} \succeq h \text{ and } f_{S \times \{t\}}^{\text{high}} f^{\text{low}} \not\cong' h,$$
 (A.29)

which contradict $\succeq = \succeq'$ in $\mathcal{T}_{\mathcal{E}}$. Therefore, it must necessarily be the case that $\bar{\pi}_S = \bar{\pi}'_S$. But then, this implies $\bar{\pi} = \bar{\pi}'$, which together with $\bar{u} = \bar{u}'$, yields

$$\succeq = \succeq' \text{ in } \mathcal{F}_{\mathcal{E}}, \tag{A.30}$$

and the proof is complete.

Proof of Theorem 3. The proof follows directly from Lemmas A1, A2 and A3. \Box

References

ALAOUI, L. & PENTA, A. (2024). What's in a u? Working Paper.

- ANDREONI, J. (1991). Reasonable doubt and the optimal magnitude of fines: should the penalty fit the crime? *RAND Journal of Economics* 22, 385–395.
- ANSCOMBE, F.J. & AUMANN, R.J. (1963). A definition of subjective probability. Annals of Mathematical Statistics 34, 199–205.
- ARROW, K. (1974). Optimal insurance and generalized deductibles. Scandinavian Actuarial Journal 1, 1–42.
- BACCELLI, J. (2017). Do bets reveal beliefs? A unified perspective on state-dependent utility issues. *Synthese* 194, 3393–3419.
- (2021). Moral hazard, the Savage framework, and state-dependent utility. *Erkenntnis* 86, 367–387.
- BAILLON, A., HUANG, Z., SELIM, A. & WAKKER, P. (2018). Measuring ambiguity attitudes for all (natural) events. *Econometrica* 86, 1839–1858.
- BENABOU, R. (2015). The economics of motivated beliefs. Revue d'Economie Politique 125, 665–685.
- BENJAMIN, D. (2019). Errors in probabilistic reasoning and judgment biases. Handbook of Behavioral Economics, Ch.2, 69–186.
- BLANCO, M., ENGELMANN, D., KOCH, A. & NORMANN, H.T. (2010). Belief elicitation in experiments: is there a hedging problem? *Experimental Economics* 13, 412–438.
- BRIER, G. (1950). Verification of forecasts expressed in terms of probability. *Monthly Weather Review* 78, 1–3.
- BULLOCK, J., GERBER, A., HILL, S. & HUBER, G. (2015). Partisan bias in factual beliefs about politics. Quarterly Journal of Political Science 10, 519–578.

- CASE, K., SHILLER, R. & THOMPSON, A. (2012). What have they been thinking? Home buyer behavior in hot and cold markets. *NBER Working Paper Series* 18400.
- CHEW, S.H. & WANG, W. (2020). On the robustness of indeterminacy in subjective probability. *Economics Letters* 188, 108939.
- COOK, P. & GRAHAM, D. (1977). The demand for insurance and protection: the case of irreplaceable commodities. *Quarterly Journal of Economics* 91, 143–156.
- DANZ, D., VESTERLUND, L. & WILSON, A. (2022). Belief elicitation and behavioral incentive compatibility. American Economic Review 112, 2851–2883.
- DE FINETTI, B. (1937). La prévision: ses lois logiques, ses sources subjectives. Annales de Ínstitut Henri Poincaré 7, 1–68.
- DELAVANDE, A. (2008). Pill, patch, or shot? Subjective expectations and birth control choice. *International Economic Review* 49, 999–1042.
- DRÈZE, J. (1961). Les fondements logiques de l'utilite cardinale et de la probabilite subjective. La Decision, Colloques Internationaux du CNRS (Paris) 73–87.

— (1987). Decision theory with moral hazard and state-dependent preferences. Essays on Economic Decisions under Uncertainty 23–89.

- DRÈZE, J. & RUSTICHINI, A. (1999). Moral hazard and conditional preferences. Journal of Mathematical Economics 31, 159–181.
- (2004). State-dependent utility and decision theory. *Handbook of Utility Theory*, Ch. 8, 839–892.
- DUCHARME, W. & DONNELL, M. (1973). Intrasubject comparison of four response modes for subjective probability assessment. Organizational Behavior and Human Performance 10, 108–117.
- EIL, D. & RAO, J. (2011). The good news-bad news effect: asymmetric processing of objective information about yourself. American Economic Journal: Microeconomics 3, 114–138.
- ENKE, B. & GRAEBER, T. (2023). Cognitive uncertainty. Quarterly Journal of Economics 138, 2021–2067.
- ESPONDA, I. & VESPA, E. (2014). Hypothetical thinking and information extraction in the laboratory. American Economic Journal: Microeconomics 6, 180–202.
- (2023). Contingent preferences and the sure-thing principle: Revisiting classic anomalies in the laboratory. *Review of Economic Studies*.
- EVERETT, J., FABER, N. & CROCKETT, M. (2015). Preferences and beliefs in ingroup favoritism. *Frontiers* in Behavioral Neuroscience 9, Art. 15.
- FEDDERSEN, T. & PESENDORFER, W. (1998). Convicting the innocent: the inferiority of unanimous jury verdicts under strategic voting. *American Political Science Review* 92, 23–35.
- FISHBURN, P. (1973). A mixture-set axiomatization of conditional subjective expected utility. *Econometrica* 41, 1–25.
- GENNAIOLI, N., MA, Y. & SHLEIFER, A. (2015). Expectations and investment. *NBER Working Paper* Series 21260.

GRANT, S. & VAN ZANDT, T. (2008). Expected Utility Theory. The Handbook of Rational and Social Choice (edited by Anand, P., Pattanaik, P. & Puppe, C.), 21–68.

GOOD, I.J. (1952). Rational decisions. Journal of the Royal Statistical Society, Series B, 14, 107–114.

- (1971). Comment on "Measuring information and uncertainty" by R.J. Buehler. *Foundations of Statistical Inference*, Toronto: Holt, Rinehart and Winston (edited by Godambe V. P. & Sprott D. A.), 337–339.
- HAMMOND, P. (1999). Subjectively expected state-independent utility on state-dependent consequence domains. *Beliefs, Interactions, and Preferences in Decision Making*, Kluwer (edited by Machina, M. & Munier, B.), 7–21.

HOSSAIN, T. & OKUI, R. (2013). The binarized scoring rule. Review of Economic Studies 80, 984–1001.

- JAFFRAY, J.Y. & KARNI, E. (1999). Elicitation of subjective probabilities when the initial endowment is unobservable. *Journal of Risk and Uncertainty* 8, 5–20.
- JEHIEL, P., MEYER-TER-VEHN, M. & MOLDOVANU, B. (2012). Locally robust implementation. *Journal* of Economic Theory 147, 2439–2452.
- JOHNSON, S., PRATT, J. & ZECKHAUSER, R. (1990). Efficiency despite mutually payoff relevant information: the finite case. *Econometrica* 58, 873–900.
- KADANE, J. & WINKLER, R. (1988). Separating probability elicitation from utilities. Journal of the American Statistical Association 83, 357–363.
- KARNI, E. (1992). Subjective probabilities and utilities with event-dependent preferences. Journal of Risk and Uncertainty 5, 107–125.
- (1993a). A definition of subjective probabilities with state-dependent preferences. *Econometrica* 61, 187–198.
- (1993b). Subjective expected utility with state-dependent preferences. *Journal of Economic Theory* 60, 428–438.

(1999). Elicitation of subjective probabilities when preferences are state-dependent. International Economic Review 40, 479–486.

- (2008a). State-dependent utility. Handbook of Rational and Social Choice, 223–238.
- (2008b). On optimal insurance in the presence of moral hazard. The Geneva Risk and Insurance Review 33, 1–18.
- (2009). A mechanism for eliciting probabilities. *Econometrica* 77, 603–606.
- —— (2011a). A theory of Bayesian decision making with action-dependent subjective probabilities. *Economic Theory* 48, 125–146.
- (2011b). Subjective probabilities on a state space. American Economic Journal: Microeconomics 3, 172–185.
- (2014). Axiomatic foundations of expected utility and subjective probability. *Handbook of the Economics of Risk and Uncertainty*, Vol. 1, 1–39.
- (2020). On the indeterminacy of the representation of beliefs by probabilities. *Economics Letters* 196, 109535.
- KARNI, E. & SAFRA, Z. (1995). The impossibility of experimental elicitation of subjective probabilities. Theory and Decision 38, 313–320.
- KARNI, E. & SCHMEIDLER, D. (2016). An expected utility theory for state-dependent preferences. Theory and Decision 81, 467–478.
- KARNI, E., SCHMEIDLER, D. & VIND, K. (1983). On state-dependent preferences and subjective probabilities. *Econometrica* 51, 1021–1031.
- KUNDA, Z. (1990). The case of motivated reasoning. Psychological Bulletin 108, 480–498.
- LOCHNER, L. (2007). Individual perceptions of the criminal justice system. *American Economic Review* 97, 444–460.
- LU, J. (2019). Bayesian identification: a theory for state-dependent utilities. *American Economic Review* 109, 3192–3228.
- MANSKI, C.F. (2004). Measuring expectations. Econometrica 72, 1329–1376.

MONONEN, L. (2023). State-dependent utility and ambiguity. Working Paper.

- MÖBIUS, M., NIEDERLE, M., NIEHAUS, P. & ROSENBLAT, T. (2022). Managing self-confidence: Theory and experimental evidence. *Management Science* 105, 504–535.
- OFFERMAN, T., SONNEMANS, J., VAN DE KUILEN, G. & WAKKER, P. (2009). A truth serum for non-

Bayesians: correcting proper scoring rules for risk attitudes. Review of Economic Studies 76, 1461–1489.

- ORTOLEVA, P. & SNOWBERG, E. (2015). Overconfidence in political behavior. *American Economic Review* 105, 504–535.
- PAUKER, S.G. & KASSIRER, J.P. (1975). Therapeutic decision making: a cost-benefit analysis. New England Journal of Medicine 293, 229–234.
- (1980). The threshold apporach to clinical decision making. New England Journal of Medicine 302, 1109–1117.
- RAMSEY, F. (1931). Truth and probability. The Foundations of Mathematics and Other Logical Essays, Wiley, NY.
- SAVAGE, L. (1954). The foundations of statistics. Wiley, NY: Dover Publications.
- (1971). Elicitation of personal probabilities and expectations. Journal of the American Statistical Association 66, 783–801.
- SCHERVISH, M., SEIDENFELD, T. & KADANE, J. (1990). State-dependent utilities. *Journal of the American Statistical Association* 85, 840–847.
- SCHLAG, K. & VAN DER WEELE, J. (2013). Eliciting probabilities, means, medians, variances and covariances without assuming risk neutrality. *Theoretical Economics Letters* 3, 38–42.
- SCHOTTER, A. & TREVINO, I. (2014). Belief elicitation in the laboratory. Annual Review of Economics 6, 103–128.
- SVENSON, O. (1981). Are we all less risky and more skillfull than our fellow drivers? Acta Psychologica 47, 143–148.
- THALER, M. (2024). The fake news effect: experimentally identifying motivated reasoning using trust in news. *American Economic Journal: Microeconomics* 16, 1–38.
- TSAKAS, E. (2017). Reasonable doubt revisited. Working Paper.
- (2019). Obvious belief elicitation. Games and Economic Behavior 118, 374–381.
- (2020). Identification of misreported beliefs. Working Paper.
- VLASTAKIS, N., DOTSIS, G. & MARKELLOS, R. (2009). How efficient is the European football betting market? Evidence from arbitrage and trading strategies. *Journal of Forecasting* 28, 426–444.
- WAKKER, P. (1989). Additive representations of preferences. Kluwer Academic Publishers.
- —— (1993). Unbounded utility for Savage's "foundations of statistics", and other models. *Mathematics of Operations Research* 18, 446–485.
- ZIMMERMANN, F. (2020). The dynamics of motivated beliefs. American Economic Review 110, 337–363.